Author's response to reviews

**Title:** Associations of body mass index and waist circumference with: energy intake and percentage energy from macronutrients, in a cohort of Australian Children

**Authors:**

Sarah A Elliott (sarah.elliott@uq.edu.au)
Helen Truby (helen.truby@monash.edu)
Amanda Lee (amanda.lee@health.qld.gov.au)
Catherine Harper (catherine.harper@health.qld.gov.au)
Rebecca A Abbott (rebecca@hms.uq.edu.au)
Peter SW Davies (ps.davies@uq.edu.au)

**Version:** 6 **Date:** 23 May 2011

**Author's response to reviews:** see over
To whom it may concern,

**RE: Manuscript number:** 7846494084434610

**Title:** “Associations of body mass index and waist circumference with: energy intake and percentage energy from macronutrients, in a cohort of Australian Children”

**Authors:** Sarah A Elliott, Helen Truby, Amanda Lee, Catherine Harper, Rebecca A Abbott, Peter SW Davies

Please find attached a point-by-point response to the matters that require attention in order for the manuscript to conform to the journal style. The authors apologise as revision 4 was made to an earlier unformatted version. The final version of the above manuscript has now been uploaded.

Yours sincerely,

Sarah Elliott  
PhD Candidate  
University of Queensland  
Children's Nutrition Research Centre  
Royal Children's Hospital  
HERSTON QLD 4029 Ph: +61 7 3365 5567  
Fax: +61 7 3346 4684  
E-mail: sarah.elliott@uq.edu.au
Major revisions:

Comment 1: Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

Authors response

This has been corrected in the submission system for all author details.

Comment 2: Abstract: This should be identical in the manuscript and the submission system.

Authors response

The abstract in the submission system is now identical to that in the manuscript.

Comments 3 and 4: Tables:

1. Tables should not be placed in situ within the body of the main text, please place them after the main manuscript text.

2. Five tables are cited within the main manuscript text, but only four tables have been included. Please correct this.

Authors response

The tables have now been moved the end of the manuscript, and the correct citations included in the main manuscript text.

Comments 5: Authors' contributions: Please indicate that all authors read and approved the final manuscript.

Authors response
A declaration has been added to the Authors contribution which states ‘All authors have read and approved the final manuscript’.

**Minor revisions:**

**Authors response**

The suggested formatting changes have been made to ensure that the manuscript now conforms to the journal style.